Author’s response to reviews

Title: The CDC blood lead reference value for children: Time for a change

Authors:

Jerome A. Paulson (jerry@envirohealthdoctor.com)
Mary Jean Brown (mjb686@outlook.com)

Version: 2 Date: 07 Feb 2019

Author’s response to reviews:

We have revised the manuscript as requested by removing citations in the abstract, revising the title page, correcting typos and list declaration, abbreviations and key words on a separate page.